Author’s response to reviews

Title: Comparison of outcomes after topography-modified refraction versus wavefront-optimized versus manifest topography-guided LASIK

Authors:

Jaeryung Kim (jaeryung502.kim@gmail.com)
Sung-Ho Choi (ophth-choi@hanmail.net)
Dong Hui Lim (ldhlse@gmail.com)
Gil-Joong Yoon (yoonalpha@gmail.com)
Tae-Young Chung (tychung@skku.edu)

Version: 1 Date: 14 Jul 2019

Author’s response to reviews:

Editor's comments:

This study aims to compare visual and refractive outcomes of three different ablation patterns in patients undergoing LASIK for myopic astigmatism.

The discussion is full of reverberations, needs to be shortened and re-organized.

-&gt; We deeply appreciate the editor for his/her constructive comment, which has undoubtedly provided us with valuable opportunities to improve our work. We stripped out needless or redundant parts in the discussion section; the length of the discussion section was reduced by three-quarters.